Author’s response to reviews

Title: Support needs of patients with obesity in primary care: a practice-list survey

Authors:

Elizabeth Evans (elizabeth.evans@newcastle.ac.uk)
Kirby Sainsbury (kirby.sainsbury@newcastle.ac.uk)
Dominika Kwasnicka (dominika.kwasnicka@curtin.edu.au)
Alex Bolster (a.bolster@newcastle.ac.uk)
Vera Araujo-Soares (vera.araujo-soares@ncl.ac.uk)
Falko Sniehotta (falko.sniehotta@newcastle.ac.uk)

Version: 1 Date: 15 Mar 2017

Author’s response to reviews:

In response to the request to revise the declarations section (copy/pasted below), I have made the requested revision. I apologise - I misinterpreted the guidelines to infer that the section should be omitted if data were not currently in a repository.

I have amended the declaration to state that the data are available from the corresponding author upon reasonable request.

Best wishes,

Dr Elizabeth Evans

________________________________________________

Declaration section:

1. Availability of Data and Materials section

Please include an “Availability of data and materials” statement section in their article detailing where the data supporting their findings can be found. BioMed Central strongly encourages all datasets on which the conclusions of the manuscript rely to be either deposited in publicly available repositories (where available and appropriate) or presented in the main paper or additional supporting files, in machine-readable format (such as spreadsheets rather than PDFs)
whenever possible. However, if you do not wish to share your data, please state that data will not be shared, and state the reason.

Availability of data and materials:  http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials

Please note that all manuscript submissions to BMC family Practice must contain a Declarations section which includes a response to each of the mandatory sub-headings listed below. Please refer to the journal’s Submission Guidelines for information regarding the criteria for each subsection

https://bmcfampract.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article